Author's response to reviews

Title: High incidence of metastatic disease in primary high grade and large extremity soft tissue sarcomas treated without chemotherapy

Authors:

Benedikt Leidinger (benedikt.leidinger@med.uni-marburg.de)
Thomas Heyse (heyse@med.uni-marburg.de)
Andreas Schuck (schuck@uni-muenster.de)
Horst Buerger (burgerh@uni-muenster.de)
Philipp Mommsen (philipp.mommsen@gmx.de)
Thomas Bruening (thomas.bruening@gmx.de)
Susanne Fuchs (fuchss@med.uni-marburg.de)
Georg Gosheger (goshegg@uni-muenster.de)

Version: 6 Date: 12 June 2006

Author's response to reviews: see over
Dear Madam or Sir,

We appreciate your positive reply concerning our paper.

Please find attached the corrected version of our manuscript considering your suggestions from 31st of May 2006. You will find the changes to be discussed in the following pages.

Sincerely yours

Benedikt Leidinger, MD

Thomas Heyse, MD
Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Authors' contributions - For the Authors' contributions we suggest the following kind of format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

Acknowledgements - Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please place this section after the Authors' contribution section.

References - The reference list should contain all authors' names, regardless of how many there are. The term 'et al.' should not be used.

Tables - Table 5 has been divided into two parts. It is important that tables are continuous, and not divided in this manner. Please either redefine the borders of the table, or rename the second section table 6. Please remember to update any changes throughout the text.

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping.

Thank you.

All points were addressed in the latest version. Please let us know, if everything is to your satisfaction.

Best wishes,

Benedikt Leidinger, MD

Thomas Heyse, MD